Author's response to reviews

Title: Effects of Microcurrent Stimulation on Hyaline Cartilage Repair in Immature Male Rats (Rattus norvegicus)

Authors:

Carla C Ciccone (camppone@hotmail.com)
Denise C Zuzzi (denise_zuzzi@yahoo.com.br)
Lia M Neves (liamara@uniararas.br)
Josué S Mendonça (posgraduacao@uniararas.br)
Paulo P Joazeiro (pjoaz@unicamp.br)
Marcelo A Esquisatto (marcelosquisatto@uniararas.br)

Version: 2 Date: 17 August 2012

Author's response to reviews:

Araras – Sao Paulo/Brazil, August 17th, 2012.

To
Professor Tom Rowles, PhD
Editor-in-Chief, BMC Complementary and Alternative Medicine
Att. Carmela S. Quidoles
Journal Editorial Office - BioMed Central

Dear Professor,

Enclosed is a revised manuscript concerning the evaluation of Effects of Microcurrent Stimulation on Hyaline Cartilage Repair in Immature Male Rats (Rattus norvegicus).

We inform that “Authors’ Contributions” and “Competing interests” sections were included. At the same time, the background section of the abstract was extended.

By the way, we reaffirm that all experiments followed indications from COBEA – Brazilian Association for Animal Experimentation and the results has not been submitted elsewhere for publication.

We look forward to hearing from you soon.

Sincerely yours,

Carla de Campos Ciccone, Denise Cristina Zuzzi, Lia Mara Grosso Neves, Josué Sampaio Mendonça, Paulo Pinto Joazeiro and Marcelo Augusto Marretto Esquisatto